Massachusetts Institute of Technology Handout 5 6.042J/18.062J: Mathematics for Computer Science February 3, 1994 Professors Charles E. Leiserson and Tom Leighton

List of students

These are exactly the students we think are taking the class: if you are on the list, we think you are taking the class; if you are not on the list, we think you are not taking the class. If you are on the list but decide to drop the class, please let us know as soon as possible.

Name Section Email address Phone number Abayan, Marlon 10 pisay92@mit.edu 541{3221 Agoyo, Acee 4 agoyo1@mit.edu 730{8212 Adler, Joseph 5 jaadler@athena.mit.edu 421{9652 Back, Heather 3 hlback@athena.mit.edu 492{6983 Bentzel, Chris 9 cbentzel@athena.mit.edu 267{4935 Boyd, Nathan 4 boydn@athena.mit.edu 267{3881 Cheiner, Oleg 10 oleg@mit.edu 225{6381 Christakos, Constantine K. 1 dean@athena.mit.edu 225{7309 Cotner, Adam 1 biff@athena.mit.edu 225{8916 Eslick, Ian 1 beethovn@ai.mit.edu x5{6124 x3{8843 Fortino, Kristen 9 kris10@athena.mit.edu 225{7305 Gladstein, Brian 5 ginzerak@mit.edu 247{3012 Gonzalez, Leo 7 leogonz@athena.mit.edu 225{6469 Gould, Brit C. 6 brit@mit.edu 731{9469 Grabowski, John 7 grubbie@athena.mit.edu 497{7922 Gupta, Parag 12 pgupta@mit.edu 225{6245 Hobart, John 11 jhobart@mit.edu 864{5398 Hooker, Chris 6 cjhooker@athena.mit.edu 225{6478 Hwang, Jungtaik 11 jthwang@mit.edu 266{2827 Ideker, Trey 4 teideker@ai.mit.edu 876{0224 Kramer, Marcus 12 marcusk@athena.mit.edu 576{9001 Lachman, Richard 7 richlach@athena.mit.edu 494{8300 Lahiri, Shounak 8 lahiri@athena.mit.edu 225{9435 Lee, A. Jennifer 2 jeni@media-lab.mit.edu 628{7489 Martinez, Jesus A. 12 salomon@athena.mit.edu 225{7318 Metzger, Michael 6 mmetzger@athena.mit.edu 225{6302 Merrill, Christopher 2 ckmerril@athena.mit.edu 262{5090

2 Handout 5: List of students

Name Section Email address Phone number Napier, Jim 1 jnapier@athena.mit.edu 661{9073 Novak, John 9 jjnovak@mit.edu 225{7333 Perez, Manuel 10 manny@mit.edu 225{9692 Phan, Lucy 8 phan@athena.mit.edu 225{8236 Preciado, Victor H. 12 supervic@mit.edu 225{8279 Ramer, Jorey 5 jramer@athena.mit.edu 536{1300 Redin, Maria 11 msredin@athena.mit.edu 864{7240 Respress, Thomas 5 tommyr@mit.edu 262{2807 Robertson, Niel 9 nielre@athena.mit.edu 494{8300 Sadowski, Mark 2 sadowski@athena.mit.edu 225{7330 Sakhuja, Rohit 4 rsakhuja@mit.edu 247{8355 Sarkar, Abhijit 4 abhijit@athena.mit.edu 225{8323 Srivastava, Amit 8 slash@mit.edu 225{7118 Toutet, Christiana 7 doidoia@mit.edu 225{7656 Van Eyk, Paul 3 vandyman@mit.edu 225{8770 Wagenberg, Becky 8 bexw@mit.edu 225{9371 Wahid, Wasiuddin 11 wasi@athena.mit.edu 225{6473 Walton, Damani 2 damani@mit.edu 247{8355 Wilson, Ely C. 3 terkaar@athena.mit.edu 225{9539 Wilson, Jason 6 jawilson@ai.mit.edu 225{6450 Yerushalmi, Yoav 3 yoav@mit.edu 225{9681 Zakharov, Alex 10 zak@mit.edu 267{0956 Zhang, Yong 5 nyker@athena.mit.edu 225{6157